HKUST 1 Computer Science

HONG KONG UNIVERSITY OF SCIENCE & TECHNOLOGY

COMP 102: Computer & Programming Fundamentals I

Summer 1996

LABORATORY ASSIGNMENT #1

1 & 3 July 1996

PROBLEM #1 Write a pr ogram which r eads in a temperatur e measur ement in degr ees Fahr enheit, converts it to degr ees Celsius , and prints the answer (in any format). Given the formula:

Example: Please enter degr ee in Fahr enheit: 77 Degree in Celsius= 25.0 Test cases: 78.8 32 -40 PROBLEM #2 Write a pr ogram to r ead in four integer values r epresenting the coor dinates of two points on a plane. Calculate the distance between the points and output the r esult with 2 decimal places. The distance can be computed by the following formula:

Example: Please enter x and y coor dinates for the first point: 3 5 Please enter x and y coor dinates for the second point: 10 2 The distance between (3, 5) and (10, 2) is: 7.62 Test cases: 3 5 10 2 10 -5 8 3 120 50 125 65

Celsius 5 9 -- Fahrenheit 32– ( ) × =

dis ce tan x_2 x_1–( )^2 y_2 y_1–( )^2 + =